Author's response to reviews

Title: A cluster randomised controlled trial of the community effectiveness of two interventions in rural Malawi to improve health care and to reduce maternal, newborn and infant mortality

Authors:

Sonia Lewycka (s.lewycka@ich.ucl.ac.uk)
Charles Mwansambo (cmwansambo@malawi.net)
Peter Kazembe (pnkazembe@baylor-malawi.org)
Tambosi Phiri (tambosi_phiri@yahoo.com)
Andrew Mganga (andymganga@yahoo.com)
Mikey Rosato (m.rosato@ich.ucl.ac.uk)
Hilda Chapota (hchapota@gmail.com)
Florida Malamba (bfloridamalamba@yahoo.com)
Stefania Vergnano (stev@doctors.org.uk)
Marie-Louise Newell (mnewell@africacentre.ac.za)
David Osrin (d.osrin@ich.ucl.ac.uk)
Anthony Costello (cihdcostello@googlemail.com)

Version: 5 Date: 9 September 2010

Author’s response to reviews: